Author's response to reviews

Title: Bio-psychosocial determinants of time lost from work following non life threatening acute orthopaedic trauma.

Authors:

Fiona J Clay (FIONA.CLAY@MUARC.MONASH.EDU.AU)
Stuart V Newstead (stuart.newstead@muarc.monash.edu.au)
Wendy L Watson (W.Watson@unsw.edu.au)
Joan Ozanne-Smith (Joan.Ozanne-smith@med.monash.edu.au)
Rod J McClure (Rod.McClure@muarc.monash.edu.au)

Version: 4 Date: 16 December 2009

Author's response to reviews:

Dear Editor

Re: Manuscript: 1984710025292378

The following revisions have been made to the article.
1. Abstract has been divided according to the headings Background, Methods, Results, Conclusion.
2. The tables have been inserted into the main body of the manuscript. In order to fit Table 1 onto one page the font size for that table has been changed to Arial 9.5.
3. The figure title and legend for figure 1 has been added to the Figure legends section immediately following the references.
4. Underlining has been removed from the body of the manuscript and been replaced with bold.
5. The short title and key words have been removed.
6. The Authors contributions have been formatted as a paragraph.

I am not sure what is meant by “Please crop the figure to minimise white space around it”. I have loaded a new version of figure 1 and hope that this is sufficient.

Yours Sincerely

Fiona Clay
Accident Research Centre
Monash University
Clayton, VIC, 3800
Australia.